Author’s response to reviews

Title: CLINICAL AND PATHOLOGICAL FACTORS INFLUENCING SURVIVAL IN A LARGE COHORT OF TRIPLE-NEGATIVE BREAST CANCER PATIENTS

Authors:

S.A.M. URRU (silvanaurru@crs4.it)
S. GALLUS (silvano.gallus@marionegri.it)
Cristina Bosetti (cristina.bosetti@marionegri.it)
Tiziana Moi (tyzumoi@yahoo.it)
Ricardo Medda (ricard@tiscali.it)
Elisabetta Sollai (elisol81@tiscali.it)
Alma Murgia (alma.murgia@aob.it)
Francesca Sanges (f.sanges@yahoo.it)
Giovanna Pira (gpira@uniss.it)
Alessandra Manca (al_manc@yahoo.it)
Dolores Palmas (dolor64@alice.it)
Matteo Floris (matteo.floris@gmail.com)
Anna Asunis (anna_asunis@yahoo.it)
Francesco Atzori (francescoatzori74@yahoo.it)
Ciriaco Carru (carru@uniss.it)
Maurizio D'Incalci (maurizio.dincalci@marionegri.it)
Massimo Ghiani (massimoghiani67@gmail.com)
Vincenzo Marras (marrasv@gmail.com)
Daniela Onnis (danielaonnis@aob.it)
Maria Cristina Santona (maricri.san@tiscali.it)
Dear Dr Rice,

please find enclosed the revised version of our manuscript, according to your requests. In particular:

1- We added the email address details for all authors on the title page

2- We checked the section "Ethics approval and consent to participate"

3- We checked that all authors have been mentioned in the "Authors' contributions"

4- We checked that the Declarations section includes all the sub-sections

5- We checked the 'Availability of data and materials' section.

Kind regards,

Cristina Bosetti